Author's response to reviews

Title: Daucus carota pentane-based fractions arrest the cell cycle and increase apoptosis in MDA-MB-231 breast cancer cells

Authors:

wassim n shebaby (wassim.shebaby@lau.edu.lb)
mohammad mroueh (mmroueh@lau.edu.lb)
kiki bodman-smith (k.bodman-smith@surrey.ac.uk)
anthony mansour (anthony.mansour@lau.edu)
robin i taleb (robin.taleb@lau.edu.lb)
costantine f daha (cdaher@lau.edu.lb)
mirvat el-sibai (mirvat.elsibai@lau.edu.lb)

Version: 2
Date: 24 February 2014

Author's response to reviews:

Please note that we have included a competing interests, an authors' contributions and an acknowledgements section as required by the editors.